Author's response to reviews

Title: Impact of newly diagnosed abnormal glucose regulation on long-term prognosis in low risk patients with ST-elevation myocardial infarction: a follow-up study.

Authors:

Eva C Knudsen (evacecilie.knudsen@ulleva.no)
Ingebjørg Seljeflot (ingebjorq.seljeflot@ulleva.no)
Michael Abdelnoor (michael.abdelnoor@ulleva.no)
Jan Eritsland (jan.eritsland@ulleva.no)
Arild Mangschau (arild.mangschau@ulleva.no)
Carl Müller (carl.muller@ulleva.no)
Harald Arnesen (harald.arnesen@ulleva.no)
Geir Ø Andersen (geiroystein.andersen@ulleva.no)

Version: 4 Date: 28 July 2011

Author's response to reviews:

The following formatting changes have been made.
1) The author list has been corrected according to style and order.
2) The trial registration is now on the submission system.
3) Figure 2 is corrected to Figure 1 B.
4) The reference list is corrected. The corrected reference list contains 30 authors.